Author's response to reviews

Title: Knee complaints seen in general practice: active sport participants versus non-sport participants

Authors:

Marienke van Middelkoop (m.vanmiddelkoop@erasusmc.nl)
Robbart LI van Linschoten (r.linschoten@erasusmc.nl)
Marjolein Berger (m.berger@erasusmc.nl)
Bart W Koes (b.koes@erasusmc.nl)
Sita MA Bierma-Zeinstra (s.bierma-zeinstra@erasusmc.nl)

Version: 4 Date: 7 March 2008

Author's response to reviews: see over
Dear editor,

We made the following formatting changes:

- Authors names listed are now identical.
- Headings are changed
- All tracking and highlighting is removed from the manuscript
- We made some Acknowledgements (see manuscript)
- The titles of the figures are now listed in the text. However, I cannot remove the legend from Figure 2, because this legend is incorporated in the figure.
- And finally, we moved all the reference citations before the full stop of a sentence.

Kind regards,

Marienke van Middelkoop
m.vanmiddelkoop@erasusmc.nl